\input{ThisWeek.tex}
\newcommand{\duedate}{Thursday, September 11}
\input{../TemplateWeek/CheckHomeWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

\begin{enumerate}
\item If you used $d$ for displacement and $k$ for the constant in being proportional, try the function $d=\sin (t\sqrt{k})$; this function shold satisfy your equation.
\item If you plug in $2$ to your answer, you should get $77e^2$.
\item If you plug in $1$ to your answer, you should get $\cos(1)$ (about $0.5403$ if you used a calculator to check).
\item If you plug $\sqrt[4]{\pi/2}$ in to your answer, you should get $4\sqrt[4]{\frac{2}{\pi}}$.
\item This is a different table:
\begin{tabular}{l|l|l|l|l}
&1&2&3&4\\
\hline
  f(x)&1&4&2&3\\
f'(x)&4&2&3&1\\
\end{tabular}
but you should get the same answer.
\end{enumerate}

\end{document}